

G OPEN ACCESS

Citation: Estrada-Franco JG, Fernández-Santos NA, Adebiyi AA, López-López MdJ, Aguilar-Durán JA, Hernández-Triana LM, et al. (2020) Vertebrate-*Aedes aegypti* and *Culex quinquefasciatus* (Diptera)-arbovirus transmission networks: Nonhuman feeding revealed by meta-barcoding and next-generation sequencing. PLoS Negl Trop Dis 14(12): e0008867. https://doi.org/10.1371/journal. pntd.0008867

Editor: Pattamaporn Kittayapong, Faculty of Science, Mahidol University, THAILAND

Received: February 24, 2020

Accepted: October 9, 2020

Published: December 31, 2020

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0008867

Copyright: © 2020 Estrada-Franco et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

RESEARCH ARTICLE

Vertebrate-Aedes aegypti and Culex quinquefasciatus (Diptera)-arbovirus transmission networks: Non-human feeding revealed by meta-barcoding and nextgeneration sequencing

José Guillermo Estrada-Franco^{1®}, Nadia A. Fernández-Santos^{1®}, Adeniran A. Adebiyi^{1®}, María de J. López-López¹, Jesús A. Aguilar-Durán¹, Luis M. Hernández-Triana², Sean W. J. Prosser¹, Paul D. N. Hebert³, Anthony R. Fooks², Gabriel L. Hamer⁴, Ling Xue⁵, Mario A. Rodríguez-Pérez¹*

 Instituto Politécnico Nacional, Centro de Biotecnología Genómica, Ciudad Reynosa, Tamaulipas, México,
Animal and Plant Health Agency, Woodham Lane, Addlestone, Surrey, United Kingdom, 3 Centre for Biodiversity Genomics, University of Guelph, Ontario, Canada, 4 Department of Entomology, Texas A&M University, College Station, Texas, United States of America, 5 College of Mathematical Sciences, Harbin Engineering University, Harbin, Heilongjiang, P.R. China

• These authors contributed equally to this work.

* drmarodriguez@hotmail.com, mrodriguez@ipn.mx

Abstract

Background

Aedes aegypti mosquito-borne viruses including Zika (ZIKV), dengue (DENV), yellow fever (YFV), and chikungunya (CHIKV) have emerged and re-emerged globally, resulting in an elevated burden of human disease. *Aedes aegypti* is found worldwide in tropical, sub-tropical, and temperate areas. The characterization of mosquito blood meals is essential to understand the transmission dynamics of mosquito-vectored pathogens.

Methodology/principal findings

Here, we report *Ae. aegypti* and *Culex quinquefasciatus* host feeding patterns and arbovirus transmission in Northern Mexico using a metabarcoding-like approach with next-generation deep sequencing technology. A total of 145 *Ae. aegypti* yielded a blood meal analysis result with 107 (73.8%) for a single vertebrate species and 38 (26.2%) for two or more. Among the single host blood meals for *Ae. aegypti*, 28.0% were from humans, 54.2% from dogs, 16.8% from cats, and 1.0% from tortoises. Among those with more than one species present, 65.9% were from humans and dogs. For *Cx. quinquefasciatus*, 388 individuals yielded information with 326 (84%) being from a single host and 63 (16.2%) being from two or more hosts. Of the single species blood meals, 77.9% were from dogs, 6.1% from chickens, 3.1% from house sparrows, 2.4% from humans, while the remaining 10.5% derived from other 12 host species. Among those which had fed on more than one species, 11% were from dogs and humans, and 89% of other host species combinations. Forage ratio analysis revealed

Data Availability Statement: All raw data has been submitted to the SRA (http://www.ncbi.nlm.nih. gov/bioproject/600226) under accession PRJNA600226.

Funding: This work was performed under the auspices of CONACyT-MEXBOL (No. 295569) to MARP; IPN (SIP 20181120 and 20195706) to MARP, and a TAMU-CONACYT Collaborative Research Grant Program (2018-041-1) to MARP and GLH. AAA holds a doctoral scholarship from Consejo Nacional de Ciencia Y Tecnología (CONACYT), Mexico (291137/457158). Funding for LMHT and ARF was provided by the Department for Environment Food and Rural Affairs (DEFRA), Scottish Government and Welsh Government through grants SV3045 and the EU Framework Horizon 2020 Innovation Grant, European Virus Archive (EVAg, grant no. 653316). Funding for JGEF was provided by IPN (SIP 20196759, 20200873, and 20202442). PNDH would like to thank Canada First Research Excellence Fund and the government of Canada through Genome Canada and the Ontario Genomics Institute to the International Barcode of Life Project which also aided the work. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

dog as the most over-utilized host by *Ae. aegypti* (= 4.3) and *Cx. quinquefasciatus* (= 5.6) and the human blood index at 39% and 4%, respectively. A total of 2,941 host-seeking female *Ae. aegypti* and 3,536 *Cx. quinquefasciatus* mosquitoes were collected in the surveyed area. Of these, 118 *Ae. aegypti* pools and 37 *Cx. quinquefasciatus* pools were screened for seven arboviruses (ZIKV, DENV 1–4, CHIKV, and West Nile virus (WNV)) using qRT-PCR and none were positive (point prevalence = 0%). The 95%-exact upper limit confidence interval was 0.07% and 0.17% for *Ae. aegypti* and *Cx. quinquefasciatus*, respectively

Conclusions/significance

The low human blood feeding rate in *Ae. aegypti*, high rate of feeding on mammals by *Cx. quinquefasciatus*, and the potential risk to transmission dynamics of arboviruses in highly urbanized areas of Northern Mexico is discussed.

Author summary

Elucidating arbovirus-vector-host contact networks is critical to understand and control mosquito-borne virus transmission, including pathogens such as ZIKV, DENV 1–4, CHIKV, and WNV. Here, we report the results of metabarcoding of blood meals of two primary pathogen mosquito vectors, *Aedes aegypti* and *Culex quinquefasciatus*. We found limited human blood feeding by *Ae. aegypti* and high preference for feeding on mammals by *Cx. quinquefasciatus*. Interestingly, blood meal analysis revealed dogs as the most utilized host for both vector species suggesting the potential for zooprophylaxis for human-amplified urban arboviruses. Pools of these vector species were tested for seven arboviruses and all were negative. We calculated vectorial capacity to discuss the potential risk and transmission dynamics of pathogens transmitted by these two important vectors in an urban location in Northern Mexico.

Introduction

Mosquitoes (Culicidae) are a diverse taxonomic group found worldwide except Antarctica [1]. Most mosquito species require a blood meal for egg maturation, and this behaviour plays a central role in the transmission of pathogens that cause millions of deaths annually. The most common arboviral vector, *Aedes aegypti* (L.) often takes more than one blood meal during a single egg maturation cycle [2], while *Culex quinquefasciatus* also transmits numerous viruses through its feeding habit. These feeding requirements elicit anthropophilic and zoophilic behaviours that result in complex blood-feeding patterns [3]. Understanding these interactions by characterizing mosquito blood meals is an important tool in unravelling host-vector relationships and associated diseases transmission dynamics [4]. Furthermore, analysing the vector host range by estimating the frequency at which vector populations feed on particular hosts can be used as a proxy for assessment of human exposure and in estimation of vectorial capacity [5].

Various techniques have been employed for the analysis of mosquito blood meals such as ELISA or precipitin tests [6,7], and PCR-based approaches [8]. Apart from the PCR-Sanger sequencing, such methods are often limited to identifying only a certain protein or region of a

gene from a specific host, rendering them amenable to only absence/presence detection of *a priori* identified blood meal hosts [3]. Even with PCR-based methods, results can be limited due to a lack of primers that amplify a wide range of vertebrate hosts [9]. Furthermore, when PCR amplification is successful, mixed-feeding events can be missed by direct Sanger sequencing [10]. High-throughput next generation sequencing technology can overcome this barrier, but to our knowledge this application has only been applied to *Anopheles* spp. [3].

The mitochondrial cytochrome *c* oxidase subunit I (COI) gene has been widely adopted for species identification because it allows species identification through the assignment of a Barcode Index Number [11,12]. It has the additional advantage that reference sequences are available for more than 700,000 animal species, far broader coverage than for other mitochondrial genes, such as *cyt b* [9] and 16S rRNA. DNA metabarcoding is a rapidly evolving method for biodiversity assessment of bulk samples using the COI gene and next-generation sequencing. It has been used in diverse applications including biodiversity monitoring, reconstruction of paleo-communities, and animal gut contents [13]. Given the importance of correctly identifying the preferred vector when examining host-vector contact [a concept expressed by the Human Blood Index (HBI)], which is defined as the proportion of insects in a vector population with human blood [14], we investigated the host feeding pattern of *Ae. aegypti* and *Cx. quinquefasciatus* using a DNA metabarcoding approach at sites near Reynosa, Mexico, a USA-Mexico border city. In addition, we tested for the presence of key arboviruses which are known to circulate in the region.

Methodology

Ethics statement

The present study was approved by the Bioethical Committee of the Faculty of Medicine at the Universidad Autónoma del Estado de México (Approval No. 011/2015) to JGEF.

Study sites and sample collection

Mosquitoes were collected at sites in Pedro J. Méndez neighbourhood (26° 1' 3.576" N, 98° 16' 30.18" W), Reynosa, Tamaulipas, in Northern Mexico (Fig 1). The neighborhood is semiurban characterized by poor infrastructure. Intrastreet roads are unpaved, and houses are mostly uncompleted or unplastered. Waste management practices are poor and streets are often flooded after rainfall. The population of the area is about 7,031 people over a *ca*. 6,016 square/meters comprising an estimated 1,750 households.

Sampling was carried out in the rainy (May to July 2018) and dry (September to November 2018) seasons in 16 households, although some received limited collections due to difficulties in access. Four traps were used for collection in each household. Host-seeking mosquitoes were collected using Biogents Sentinel 1 traps (Biogents AG, Germany); one trap was placed outside each home on a weekly basis for a 24-hour period. Gravid mosquitoes were collected using the Autocidal Gravid Ovitraps (AGO) traps (Springstar) that were checked weekly [15]. Indoor resting and host-seeking populations were collected using a CDC backpack aspirator (John W. Hock Company, No. 1412.01, Gainesville, FL) for about 3-min; outdoor resting and blood-engorged mosquitoes were collected using the back-pack aspirator that collected all resting mosquitoes inside a 1 m³ wooden box opened on one side with both the outside and the inside painted red. Collected mosquitoes were preserved on ice in the field and transported to the laboratory, where they were stored at -80°C until processing. Sample identification and sorting were carried out on a BioQuip chill plate (BioQuip, USA). Mosquito morphological identification was conducted using standard keys [1,16]. Female mosquitoes were further classified into different feeding stages and only engorged mosquitoes of Sella scale 2 (freshly fed)



99°0'0"W

Fig 1. Study site and location of traps in Pedro J. Mendez neighbourhood in Reynosa, Northern Mexico. Fig 1 is an original map, created by co-author AAA and is available for use under the CC-BY license. ESRI 2011. ArcGIS Desktop: Release 10. Redlands, CA: Environmental Systems Research Institute.

https://doi.org/10.1371/journal.pntd.0008867.g001

and 3 (late stage fed) were processed for the blood meal analysis [17]. Engorged mosquitoes were placed individually into nuclease-free 1.5 mL micro-centrifuge tubes, labelled with an identification number and stored at -80° C for further processing.

The study area was residential and the objective of the study is to understand vector behavior in this residential area. The household selection was random and largely dependent on permission granted by house owners as well as availability to allow entrance on their properties to check traps and retrieved mosquitos. Mosquito collection was done outdoors in the yard of these properties using AGO traps, Biogents Sentinel 1 traps (BGS1), and the resting boxes. Only the backpack aspiration was done indoors. Also, the nature of the study area, where all of the houses shared fences, provided opportunity to sample mosquitoes within a 50-meter radius of each outdoor trap, as this is within the flight range of mosquitoes. Furthermore, with a sampling duration of 5 months, we have a strong reason to believe there is enough representation of the mosquito population in this area. We did not attempt to quantify the behavior of the

household occupants and how much time was spent inside versus outside. However, anecdotally, we observed a significant proportion of their activities, including cooking, grilling, and evening family gatherings, were done outside. Permeability of houses to mosquitoes as well was not measured. Given mosquitoes can have a flight range of up to 100 m post-feeding [18] and species of *Aedes* and *Culex* have 2–5 km flight range [19], we do not think the spatial distribution of the households might have significantly impacted the bloodmeal feeding pattern observed.

DNA extraction and purification

Blood-fed Ae. aegypti and Cx. quinquefasciatus were sent to the Canadian Centre for DNA Barcoding, Centre for Biodiversity Genomics, University of Guelph (Guelph, ON, Canada) for DNA extraction and sequencing. The frozen mosquitoes were incubated in liquid nitrogen for one minute to ensure that the blood meal did not thaw and leak during removal of the abdomen. The abdomens were removed with forceps and placed directly into 60 µL of lysis buffer [700 mM guanidine thiocyanate (Sigma), 30 mM EDTA pH 8.0 (Fisher Scientific), 30 mM Tris-HCl pH 8.0 (Sigma), 0.5% Triton X-100 (Sigma), 5% Tween-20 (Fluka Analytical), and 2 mg/mL of Proteinase K (Promega)], which was then incubated for 18 h at 56°C with gentle shaking. DNA purification employed the glass fiber technique described by Ivanova et al. [20]. The lysate was mixed with two volumes (120 μ L) of binding mix [3 M guanidine thiocyanate, 10 mM EDTA pH 8.0, 5 mM Tris-HCl pH 6.4, 2% TritonX-100, and 50% ethanol] and applied to a well of a 96-well silica membrane plate (PALL), 700 µL at a time. The plate was centrifuged at 5000g for 5 min after each application. The membrane was washed once with 180 µL of protein wash buffer [1.56 M guanidine thiocyanate, 5.2 mM EDTA pH 8.0, 2.6 mM Tris-HCl pH 6.4, 1.04% Triton X-100, and 70% ethanol] and centrifuged at 5000g for 2 min. The plate was then washed a second time with 750 µL of wash buffer [50 mM NaCl (Fisher Scientific), 0.5 mM EDTA pH 8.0, 10 mM Tris-HCl pH 7.4, and 60% ethanol] and centrifuged at 5000g for 5 min. Centrifugation was repeated a second time to remove residual wash buffer. All traces of wash buffer were removed by incubating the plate at 56°C for 30 min. To release DNA from the silica membrane, 40 µL of elution buffer [10 mM Tris-HCl, pH 8.0, pre-warmed to 56°C] was applied directly to the membrane and allowed to incubate at room temperature for 1 min. DNA was eluted from the membrane (into a clean plate) via centrifugation at 5000g for 5 min. A negative control (lysis buffer only) was processed in parallel with the samples.

PCR amplification and sequencing

Host DNA was prepared for deep sequencing using a two-stage fusion primer approach [13] with vertebrate-specific primers. Two microliters of DNA extract was added to a PCR reaction consisting of 6.25 μ L of 10% D-(+)-trehalose dihydrate (Fluka Analytical), 2.0 μ L of Hyclone ultra-pure water (Thermo Scientific), 1.25 μ L of 10X Platinum Taq buffer (Invitrogen), 0.625 μ L of 50 mM MgCl2 (Invitrogen), 0.125 μ L of each 10 μ M primer, 0.0625 μ L of 10 mM dNTP (KAPA Biosystems), and 0.060 μ L of 5U/lL Platinum Taq DNA Polymerase (Invitrogen) for a total reaction volume of 12.5 μ L. The primers were a cocktail of BloodmealF1_t1, BloodmealF2_t1, VR1_t1, VR1d_t1, and VR1i_t1 (S1 Table), which were tailed with M13F and M13R sequences and targeted a 185 bp fragment of the COI barcode region. The BloodmealF primers were designed in this study to amplify DNA from common vertebrates used by mosquitoes, not to amplify insect DNA including mosquito, and to amplify short fragments (<200 bp) given the nature of degraded DNA in blood meals. Thermocycling consisted of an initial denaturation at 95°C for 2 minutes, 60 cycles of 95°C for 40 seconds, 56°C for 40 seconds, and 72°C for 30 seconds, and a final extension at 72°C for 5 minutes. PCR products

were visualized on a 2% E-gel (Invitrogen) to confirm amplification, and then diluted two-fold with sterile water.

The diluted products were then used as a template for a second round of PCR using M13F primers tailed with Ion Xpress universal molecular identifier (UMI) tags and the Ion Torrent "A" sequencing adapter, and M13R primers tailed with the Ion Torrent trP1 sequencing adapter (S1 Table). Reaction components for the second round of PCR were identical to the first and thermocycling consisted of an initial denaturation at 95°C for 2 minutes, 5 cycles of 95°C for 40 seconds, 45°C for 40 seconds, and 72°C for 30 seconds, 35 cycles of 95°C for 40 seconds, 51°C for 40 seconds, and 72°C for 30 seconds, and a final extension at 72°C for 5 minutes. The products from the second round of PCR were pooled in equal volumes and purified by mixing 400 μ L of pooled product with 400 μ L of purification beads (Aline Biosciences, Woburn, MA, USA) for a ratio (v:v) of 1:1 (beads:DNA). The mixture was incubated at room temperature for eight minutes to allow the DNA to bind to the beads, after which the beads were pelleted on a magnet. The supernatant was discarded, and the pellet was washed three times with 1 mL of freshly prepared 80% ethanol and then air dried for 10 minutes. The purified product was eluted from the beads by resuspending the dried pellet in 200 μ L of sterile water, pelleting the beads, then carefully transferring 180 μ L of the supernatant to a clean 1.5 mL tube. The purified product was quantified using a Qubit 2.0 fluorometer and adjusted to 26 pM with sterile water. The 26 pM library was prepared for sequencing and loaded onto a 530-chip using an Ion Chef automated platform (Thermo Scientific) with standard 400 bp chemistry. Deep sequencing was performed on an Ion Torrent S5 using 850 flows.

Bioinformatics analysis

Following sequencing, raw sequence reads were automatically assigned to samples (via the UMI tags) by the Torrent Browser suite. The reads were filtered based on a minimum quality value of QV20 and a minimum read length of 225 bp using Sickle (github.com/ucdavisbioinformatics/sickle). Adapter and primer sequences were removed using CutAdapt [21]. Since the forward primer should be readily visible in the reads, any reads in which the forward primer was not recognized were discarded to ensure that only high-quality reads were included in the final dataset. Additionally, trimmed reads were only retained if they were between 150-200 bp. Reads passing these filters were collapsed into unique sequences (http:// hannonlab.cshl.edu/fastx toolkit/index.html) while retaining original read counts. The collapsed sequences were then clustered into operational taxonomic units (OTUs) using 97% identity [22], and OTU sequences were retained only if they were less than 220 bp in length and composed of at least 100 reads. Each OTU sequence was used to query (BLAST) a custom database composed of global vertebrate COI sequences downloaded from BOLD. The resulting BLAST hits were filtered to retain only those with a minimum match of 95% identity and 100 bp of coverage between the queried sequence and a reference sequence. Furthermore, identifications were only considered reliable if supported by at least 50 original reads. All raw data has been submitted to the SRA under accession PRJNA600226.

Animal surveys

In order to estimate the relative abundance of avian species in the trapping areas, a bird survey was conducted as described by Brugman [17] in eight points approximately 50 meters apart between 08:00 and 10:00 hrs. when the avian population will still be relatively more active. Observations were made at each point for 10 minutes, and other observed vertebrates, including horse and dogs, were also noted. We also conducted individual household censuses of vertebrates in the trapping area using a questionnaire. An adult in each household was asked for

the number of dogs, cats, chickens, and other animals in the household. Where possible, study personnel conducted headcounts of the animals for each household. The combined total number of animals for each species recorded was used to estimate the abundance of common vertebrates in the trapping area [17].

Viral extraction and detection by qRT-PCR

Non-engorged Ae. aegypti mosquitoes were tested for arboviruses including the four serotypes of DENV, ZIKV, and CHIKV. Mosquito pools were processed according to standard procedures [23]. In brief, each mosquito pool was homogenized in 600 µL of Hank's buffer solution (Gibco cat. #14170112) with a single bead and centrifuged for 5 minutes at 18000 g. RNA extraction was carried out using 100 µL of the supernatant with a MagMax CORE Nucleic Acid Purification Kit (ThermoFisher, Waltham, MA) following the manufacturer's instructions. For a subset of samples we quantified nucleic acid recovered using a spectrophotometer (Epoch, BioTek Instruments, Inc.). DENV was tested for using the PathID Multiplex one step (Applied Biosystems) and Qiagen QuantiTect probe reverse transcriptase polymerase chain reaction (RT-PCR) kits with the primers DENV_F, DENV_R1-3, DENV_R4, and DENV_P probe described by Alm et al. [23]. DENV strains used as positive controls were provided by the University of Texas Medical Branch. We added 5 µL of RNA to a 20 µL master mix containing 12.5 µL RT-PCR buffer, 0.1125 µL each of the primers, 0.2 µL of DENV_P probe, 0.25 µL enzyme, and RNase free water. Cycling conditions were as follows: reverse transcription cycle at 50°C for 30 minutes, followed by an RT-enzyme inactivation cycle at 95°C for 20 minutes, then 45 cycles of 95°C for 3 seconds and 60°C for 30 seconds. CHIKV and ZIKV were tested for using a multiplex RT-qPCR described by Lanciotti et al. [24,25]. Briefly, 5 µL of RNA was added to a master mix containing $0.3 \,\mu\text{M}$ of each probe, $2 \,\mu\text{L}$ of each of the four primers, 2.5 µL of enzyme mix and RNase free water. Zika PRVABC59 RNA lysate and CHIKV R80422a [23] RNA lysate received from the CDC were used as positive controls. Cycling conditions were as follows: reverse transcription cycle at 48°C for 10 minutes, followed by an enzyme inactivation cycle at 95°C for 10 minutes, then 40 cycles of 95°C for 15 seconds, and 60°C for 1 minute. All samples with Ct values >38 were considered negative [22,23]. Cx. quinquefasciatus was tested for WNV using the primers WNENV-forward (TCAGCGATCTC TCCACCAAAG) and WNENV-reverse (GGGTCAGCACGTTTGTCA-TTG) and the probe WNENV-probe as described by Hamer et al. [26].

Data analysis

We estimated the forage ratio (FR), defined as the frequency at which a mosquito selects a vertebrate host over other available vertebrate hosts, by dividing the observed frequency of blood meals by the expected frequency of blood meals of a given species [27,28]. FR = s/a where s = the percentage of female mosquitoes containing blood of a particular host, and a = percentage of the total available host population represented by that particular host (as determined by the animal survey). A FR of 1.0 indicates mosquitoes are feeding on hosts in equal proportion to availability, whereas values >1.0 indicate over-utilization and values <1.0 indicate under-utilization. The human blood index (HBI) was calculated as the sum of human blood meals for a mosquito species divided by the sum of engorged mosquitoes of a particular species with successful host identification [29].

Virus screening results were analysed using the Bayesian algorithm in the Katholi's poolscreen program as previously described [30]. The 95% exact upper limit confidence interval (95%-ULCI) for point prevalence of arbovirus infection in both vector species populations were estimated.

Mathematical modeling

Investigating the transmission efficiency and risk of human exposure to *Ae. aegypti*-transmitted viruses are essential steps to developing successful intervention strategies. Thus, vectorial capacity (VC) that incorporates blood feeding behaviour and vector longevity along with vector competence was calculated to determine the intensity of dengue virus transmission with the following equation [31]:

$$VC = \frac{h \cdot b \cdot p^n}{-\ln(p)}$$

where n = extrinsic incubation period; b = vector competence; p = survival rate; and h = blood feeding rate.

We applied the model of Chan and Johansson [32], which relates the extrinsic incubation period (EIP) with temperature as a covariate. The EIP (*n*, in days) was estimated as a function of temperature (*T*) using the following equation: $n(T) = 600(0.3/2\pi)^{0.5} \exp\left(\frac{-0.3(T-5.9)^2}{\pi}\right)$.

Vector competence (b) was calculated by using the equation of Sallum *et al.* [33]

$$b = \frac{\text{Incidence}(t)}{aS_{H}(t)} \frac{N_{H}(t)}{N_{M}(t)} \frac{N_{M}(t)}{I_{M}(t)}$$

where Incidence (t) is the number of new dengue cases per unit time, $S_H(t)$ is the number of susceptible humans, $N_M(t)/N_H(t)$ is the ratio of the number of dengue mosquitoes to the number of humans, and $I_M(t)/N_M(t)$ is the prevalence of dengue virus infection in the vector population.

The survival rate, *p*, can be expressed by $e^{-\mu}$ [34]. Besides, according to the experiments by Yang *et al.* [35] in female *Ae. aegypti* over the temperature range of 10.54°C< = T< = 33.41°C, a 4th order polynomial function can be used to describe the relationships between the temperature and μ as follows:

$$\mu(T) = 0.8692 - 0.1590T + 0.01116T^2 - 3.408x10^{-4}T^3 + 3.809x10^{-6}T^4.$$

Results

Mosquito relative abundance

Mosquito samples were collected a total of 423 times, with each collection event involving checking all four traps during a visit. Empty traps were not recorded. A total of 15,922 individual Ae. aegypti and Cx. quinquefasciatus mosquitoes were collected (13,761 from May-July and 2,161 from September-November 2018; S2 Table). A total of 881 individual mosquitoes of other species were also collected which summed to 16,733 individuals of all mosquitoes combined and for all trap types. The taxonomic composition included 13 identified species. The most abundant species of mosquitoes were Ae. aegypti (n = 5,211; 31.1%) and Cx. quinquefas*ciatus* (n = 10,711; 64.0%; S2 Table). The remaining 11 species of mosquitoes were as follows: *Ae. albopictus* (n = 70; 0.41%), *Aedes* spp. (n = 3; 0.02%), *Anopheles pseudopunctipennis* (n = 3; 0.02%), *Anopheles pseudop* (0.02%), Cx. salinarius (n = 3; 0.02\%), Cx. tarsalis (n = 3; 0.02\%), Cx. nigripalpus (n = 7; 0.04\%), *Cx. interrogator* (n = 24; 0.14%), *Cx. restuans* (n = 59; 0.35%), *Cx. erythrothorax* (n = 6; 0.03%), Ae. bimaculatus (n = 1; 0.005%), and An. quadrimaculatus (n = 2; 0.01%). The remaining 3.83% (n = 670) of the mosquitoes were unable to be identified morphologically due to lack of identifiable features or damage body parts. Forty percent (n = 6,477) of the Ae. aegypti and Cx. quinquefasciatus mosquitoes collected were females and 60% (n = 9,445) were males (S2 Table). The mean number of Ae. aegypti and Cx. quinquefasciatus mosquitoes collected from

the BGS1 traps per collection was 82.83 (\pm 11.97 SEM; 95%-CIs = 59.1–105.5), significantly higher than any other trap type (Table 1).

Blood feeding pattern

A total number of 1,044 bloodfed mosquitoes were collected which included 436 Ae. aegypti (41.8%) and 608 Cx. quinquefasciatus (58.2%). Overall, 170 Ae. aegypti (3 collected using backpack aspirators, 52 collected using BGS1 traps, and 115 collected from resting boxes) and 475 Cx. quinquefasciatus (2 collected from AGO trap, 10 collected from CDC backpack aspirators, 269 from BGS1 traps, and 194 from resting boxes) were used in the blood meal analysis. Of the 170 Ae. aegypti individual mosquitoes examined, 145 (85%) yielded host identifications. Of the 145 successful individuals, 107 (73.8%) exhibited single host feeding and 38 (26.2%) exhibited mixed host feeding (*i.e.* more than one vertebrate species). Of the 107 individuals that fed on a single host, 54.2% (n = 58) fed on dogs (*Canis lupus familiaris*), 28.0% (n = 30) fed on humans (*Homo sapiens*), 16.8% (n = 18) fed on cats (*Felis silvestris sil*vestris), and 1.0% (n = 1) fed on red-eared turtle (*Trachemys scripta*) (Fig 2). Of the 38 individuals that fed on mixed hosts, 65.9% (n = 25) fed on dogs and human, 18.4% (n = 7) fed on dogs and cats, 5.3% (n = 2) fed on dogs and cattle, and 2.6% (n = 4) fed on other host combinations (Fig 3 and S3 Table). There was no significant difference (p = 0.9989) in the host identification success between the two Sella scores used: 85.3% with Sella score 2 and 85.2% with Sella score 3.

A lower sequencing success was achieved for *Cx. quinquefasciatus*. Out of the 475 individuals examined, 388 (81.7%) yielded host identifications. Of the 388 individuals, 326 (84.0%) exhibited single host feeding, while 63 (16.2%) exhibited mixed host feeding. Of the 326 individuals that fed on a single host, 77.9% (n = 254) fed on dogs, 6.13% (n = 20) fed on chickens (*Gallus gallus*), 3.1% (n = 10) fed on house sparrows (*Passer domesticus*), 2.4% (n = 8) fed on humans, 2.1% (n = 7) fed on Eurasian collared doves (*Streptopelia decaocto*), 2.1% (n = 7) fed on Virginia opossums (*Didelphis virginiana*), 2.1% (n = 7) fed on mourning dove (*Zenaida macroura*), 0.9% (n = 3) fed on Inca dove (*Columbina inca*), 0.6% (n = 2) fed each on cats and rock dove (*Columba livia*), and the remaining five *Cx. quinquefasciatus* fed on a wide range of vertebrate species (n = 20), with the most frequent combinations being dog and human (11.1%, n = 7), dog and house sparrow (9.5%, n = 6), and dog and turkey (*Meleagris gallopavo*; 9.5%, n = 6; Fig 3 and S3 Table). There was no significant difference (p = 0.260) in host identification success between the two Sella score used: 82.5% with Sella score 2 and 77% with Sella score 3.

		Resting red box	CDC backpack aspirator	BG Sentinel trap	AGO trap	
		$\underline{x} \pm SEM$	$\underline{x} \pm SEM$	$\underline{x} \pm \text{SEM}$	$\underline{x} \pm \text{SEM}$	
Aedes aegypti	Male	5.42 ± 1.45 (2.55-8.28)	0.90 ± 0.49 (-0.10-1.89)	10.89 ± 1.86 (7.23-14.56)	0.24 ± 0.05 (0.14-0.34)	
	Female	4.25 ± 0.56 (3.13-5.37)	2.31 ± 0.50 (1.28-3.34)	$14.14 \pm 1.80 (10.58 - 17.69)$	2.80 ± 0.25 (2.31-3.28)	
Culex quinquefasciatus	Male	15.49 ± 3.87 (7.87-23.14)	1.17 ± 0.82 (-0.51-2.85)	36.37 ± 7.15 (22.24-50.50)	0.65 ± 0.14 (0.37-0.92)	
	Female	6.65 ± 1.38 (3.93-9.38)	1.52 ± 0.43 (0.63-2.41)	16.18 ± 3.32 (9.62-22.74)	3.01 ± 0.47 (2.09-3.93)	
Overall		32.56 ± 5.86 (20.96–44.16)	6.07 ± 1.66 (2.67–9.47)	82.83 ± 11.97 (59.15-105.50)	6.82 ± 0.60 (5.63-8.01)	

Table 1. Average ± SEM of mosquitoes per trap per collection. The total mosquito counts retrieved from each trap were averaged by the number of collection events that yielded the count, which translates to the number of individuals per trap per collection event.

Values in parentheses represent the 95%-confidence intervals.

https://doi.org/10.1371/journal.pntd.0008867.t001



Fig 2. Host identification for single blood meals from *Aedes aegypti* and *Culex quinquefasciatus*. *Ae. aegypti* = 25 (14.7%) failed blood meal identification, but 145 individuals were examined; *Cx. quinquefasciatus* = 87 (18.3%) failed identification, but 388 individuals were examined. (Figure percentage is for successful identification).

https://doi.org/10.1371/journal.pntd.0008867.g002

Forage ratio and human blood index

Based on our animal survey, each household contained an average of 3.6 ± 0.45 SEM) human inhabitants, and the study area contained an average of 79 humans, 65 dogs, 48 cats, 37 chickens, and 3 pigs (S4 Table). Forage ratios for *Ae. aegypti* and *Cx. quinquefasciatus* were calculated with host availability estimated from these results. The highest forage ratio of *Ae. aegypti* was for dogs (4.34), which was nearly twice as high as the second highest forage ratio for humans (2.16; Table 2). The highest forage ratio of *Cx. quinquefasciatus* was also for dogs (5.64), followed by Virginia opossum (*Didelphis virginiana*; 4.83) and turkey (4.83; Table 3). The human blood index (total number of human blood meals divided by the total number of engorged females with a confirmed result) for *Ae. aegypti* and *Cx. quinquefasciatus* was 39.3% and 4.1%, respectively.

Mosquito arbovirus RNA testing

A total of 2,941 female *Ae. aegypti* were collected during the study. Out of these, 118 pools were made from 2,545 mosquitoes with an average of 18.4 (\pm 1.73 SEM) mosquitoes per pool. These were tested and none was found positive (point prevalence = 0) for DENV 1–4, ZIKV and CHIKV using qRT-PCR (positive controls returned positive results, indicating that the assay functioned as expected). Similarly, of the 3,536 female *Cx. quinquefasciatus* specimens, 37 pools, representing 1,057 mosquitoes, was tested for WNV (using qRT-PCR), and all were found to be negative (point prevalence = 0). The associated 95%-ULCI for prevalence of any of



Fig 3. Host identification for mixed blood meals from *Aedes aegypti* and *Culex quinquefasciatus*. *Ae. aegypti* = 25 (14.7%) failed blood meal identification, but 145 individuals were examined; *Cx. quinquefasciatus* = 87 (18.3%) failed identification, but 388 individuals were examined. (Figure percentage is for successful identification).

https://doi.org/10.1371/journal.pntd.0008867.g003

the viruses examined in *Ae. aegypti* was 0.75/1,000 (0.075%) mosquitoes examined, while the associated 95%-ULCI for prevalence of WNV in *Cx. quinquefasciatus* was 1.79/1,000 (0.18%) mosquitoes examined (Table 4).

Table 2. Forage ratio estimates for Aedes aegypti collected in Reynosa, Northern Mexico.

Animal host	Number of hosts blood meal identified	% host fed	Estimated host abundance	Forage Ratio	HBI
Dog	94	51.37%	11.84%	4.34	
Human	57	31.15%	14.39%	2.16	39.31%
Cat	26	14.21%	8.74%	1.63	
Cow	2	1.09%	-	n/a	
Horse	1	0.55%	1.28%	0.43	
Muscovy Duck	1	0.55%	0.18%	3	
Tortoise	1	0.55%	0.55%	1	
Chicken	1	0.55%	6.74%	0.08	
Overall	183*	100.00%			

*This is greater than the 145 number of Ae. aegypti with successful host identification because each mixed feeding event represented two host species.

n/a = not surveyed.

HBI-Human blood proportion or engorged females containing human blood.

https://doi.org/10.1371/journal.pntd.0008867.t002

Vertebrate host	# of host fed	% of host fed	Estimated abundance	Forage ratio
Dog	304	66.81%	11.84%	5.64
Chicken	31	6.81%	6.74%	1.01
House sparrow	18	3.96%	18.40%	0.22
Human	16	3.52%	14.39%	0.24
Mourning dove	13	2.86%	0.73%	3.92
Virginia opossum	12	2.64%	0.55%	4.83
Eurasian collared dove	10	2.20%	4.37%	0.50
Cat	8	1.76%	8.74%	0.20
Rock dove	8	1.76%	1.09%	1.61
Turkey	8	1.76%	0.36%	4.83
Inca dove	6	1.32%	15.30%	0.09
Mallard	4	0.89%	0.36%	2.47
Andean opossum	3	0.66%		
Horse	3	0.66%	1.28%	0.52
Bewick's wren	3	0.66%		
Cow	2	0.44%		
Pig	2	0.44%	0.55%	0.80
Texas tortoise	1	0.22%	0.55%	0.40
Northern yellow bat	1	0.22%		
Nicaraguan grackle	1	0.22%		
Striated heron	1	0.22%		
Overall	455*	100.00%		

Table 3. Forage ratio for *Culex quinquefasciatus* collected in Reynosa, Northern Mexico.

*This is greater than the 388 number of *Cx. quinquefasciatus* with successful host identification because of mixed host feeding, with each mixed feeding representing two host species.

https://doi.org/10.1371/journal.pntd.0008867.t003

Mathematical modelling

According to the findings of Rodríguez-Pérez *et al.* [36], six people were IgM-positive and seroconverted from a total of 77 examined people, therefore the Incidence/S_H = 6/(77–6). The prevalence of dengue virus infection should be varying with time. We assumed that it is a constant due to the absence of such data. Similarly, the ratio of the number of new cases to the number of humans is also assumed to be constant. We derived the ratio of the number of humans to the number of mosquitoes. Namely, 1,527 landing counts in 231 households; then assuming on average four persons per household, yields $N_H/N_M = 231^*4/1527$. The infection rate of DENV in *Ae. aegypti* is 0.47% (3/633) as from Rodríguez-Pérez *et al.* [36]. The biting rate (*a*) as from Garza-Hernández *et al.* [37], where a = 1/5, the human biting rate for DENV in Mexico.

Table 4. qRT-PCR virus screening results of the mosquito pools examined.

Mosquito species	Virus screened	Number of pools examined	Pool size range	Number of mosquitoes	qRT-PCR Positive pools	Point prevalence of infected mosquitoes/10,000*
Ae. aegypti	DENV (1–4), ZIKV, CHIKV	118	1-83	2,545	0	0 (7.51)
Cx. quinquefasciatus	WNV	37	1–74	1,057	0	0 (17.96)

*The value represents the maximum likelihood point estimate and the value in parentheses represents the upper bound 95%-confidence interval.

https://doi.org/10.1371/journal.pntd.0008867.t004

Furthermore, the blood feeding rate *h* be (105/1106) * (57/145) = 3.7% (S2 Table). Here, 105 is the number of fed females and 1,106 is the total caught number of *Ae. aegypti*. According to Figs 2 and 3, 57/145 = 39.3% is the proportion of human blood meals identified. Then 105*57/145 is the number of mosquitoes that will bite human among the total 1,106 mosquitoes. Therefore, 3.7% is the proportion of mosquitoes that will bite humans among the total mosquito population.

Using the equation of vectorial capacity, we estimated VC of *Ae. aegypti* for DENV in Reynosa which ranged from 112 to 172 from temperatures ranging between 22 and 30°C, respectively (Fig 4).

Discussion

Information on vector-host interactions and associated disease transmission dynamics obtained through the identification of blood meal origin in mosquitoes is important for disease control [4].



Fig 4. The relationship between temperature and vectorial capacity of *Aedes aegypti*. Extrinsic incubation period, vector competence, survival rate, and vectorial capacity were calculated using models as reported elsewhere [31–34]. It should be clarify that the vectorial capacity model used is for the mosquito population in general, instead of an overall model of vectorial capacity for *Aedes aegypti*.

https://doi.org/10.1371/journal.pntd.0008867.g004

The acquisition of a blood meal is an essential component of the vectorial potential of any mosquito vector. Also important is whether the female mosquito preferentially feeds on animals or humans, particularly in estimating the basic reproduction number associated with the transmission potential of the system under study. This, in turn, is closely linked to vector control strategies. In this context, the occurrence of epidemics involving viruses transmitted by *Ae. aegypti* have increased in the Americas. In the present study, we observed an overall relatively low density for *Ae. aegypti* in residential neighbourhoods of Reynosa, relative to *Cx. quinquefasciatus*. An earlier study in the Nuevo Amanecer neighbourhood in the eastern tip of Reynosa city, reported five host-seeking *Ae. aegypti* per household [36], a stark contrast to 20 females/household reported in a dengue-endemic tropical country of Thailand [38]. Although it is arguable that a low abundance of *Ae. aegypti* might reduce disease transmission, there is limited knowledge as to how this influences the ability of *Ae. aegypti* to feed on humans in highly urbanized areas of Northern Mexico, and how proportional human feeding might drive virus transmission potential.

Aedes aegypti is considered a highly anthropophilic species [39–42]; however, we found that 61.7% (129/209 examined; <u>S3 Table</u>) of individual *Ae. aegypti* with a blood meal result had fed on non-human hosts. A similar study conducted on the opposite side of the US-Mexico border in South Texas recently found a similar result, with 70% of *Ae. aegypti* feeding on non-human hosts [43]. In the current study, the proportional feeding of *Ae. aegypti* on domestic dogs (54.2%) was almost twice that of the second most common host, humans (28.0%). This was also similar to the findings of the study north of the US-Mexico border where dogs were responsible for 50% of the *Ae. aegypti* host identifications [43]. However, the blood meal analysis methodology of the Olson *et al.* [43] study was PCR-Sanger sequencing which is different to the meta-barcoding next-generation sequencing approach used in the current study, making the results difficult to compare.

Equally surprisingly is the host feeding pattern observed in *Cx. quinquefasciatus*, with domestic dog as the dominant host species. This finding was unexpected because prior studies have shown that *Cx. quinquefasciatus* is strongly ornithophilic [8,44]. Estimating the relative density of wild and domestic animals is difficult, and although our study found that chicken was the second most fed on host by proportion, forage ratio analysis revealed dog, Virginia opossum, and turkey as the most over-utilized hosts. The domestic dog was over-utilized by Cx. quinquefasciatus even more so than Ae. aegypti, with a 20-fold (79.2%) increase over the second most abundant host, chicken (3.7%). Our forage ratio analysis was based on a survey of animals present in these communities. Given that we were attempting to sample a combination of wild and domestic animals, we used a combination of a 'census', by counting all animals observed during point counts as well as a 'survey', in which a subsample of the animals present was estimated. We asked the homeowners for the number of animals in each household. The forage ratios are based on the relative abundance of these vertebrates, which would be difficult to compare wild animals (such as birds) to domestic animals (such as dogs) given the difference in survey methodologies. The interpretation of the high forage ratio on dogs is difficult to interpret for this reason. Prior published studies report that *Culex quinquefasciatus* are ornithophilic [45] so the difference compared to the current study could be due to host availability. The Olson et al. [43] study conducted in South Texas found that only 22% of the Cx. quinquefasciatus blood meals came from dogs while 67% were from chicken and 9% were from wild birds. In the South Texas low-income residential study sites, dogs made up 31% of the vertebrates surveyed in the neighborhood and chickens made up 17% [43]. In the current study, dogs accounted for 11.8% of the vertebrates and chickens accounted for 6.7%, although the survey in the current study included wild birds which were not included in the South Texas study. Considering the slightly different survey designs, the availability of dogs versus chickens appear similar in the low-income communities north and south of the US-Mexico border.

Mixed feeding patterns were also observed in common hosts: 57 (39.3%) Ae. aegypti out of 145 individuals examined fed on humans (30 individuals contained single feeding with human blood and 27 individuals contained mixed feeding with human blood; Figs 2 and 3). Very limited studies have reported a feeding pattern in Ae. aegypti where humans represent less than half of the blood meals (but see [34]). To our knowledge, this is the first study using next-generation sequencing for the blood meal analysis of Ae. aegypti, making it difficult to compare these results to prior published studies. Prior studies on Ae. aegypti blood meal analysis often utilized precipitin or ELISA assays, however, ELISA has been reported to have less specificity with assays involving human, dog, and cat blood meals [46]. Additionally, PCR and direct sequencing of the amplicon could miss blood meals containing more than one vertebrate species. Therefore, previous studies likely underestimated the frequency of Ae. aegypti feeding on multiple vertebrate species. Additionally, multiple feeding on different individuals of the same species likely occurs frequently, but our methodology would not have detected these observations. Although relatively still expensive in resource-limited settings, advances in next generation sequencing techniques and higher taxonomic coverage in reference databases have enabled the exploration of the meta-barcoding approach for a more reliable analysis of mosquito blood meals.

Anopheles gambiae was shown to exhibit polymorphism for host preference [47], and histological examination revealed that nearly one-half of all engorged *Ae. aegypti* females in Thailand contained more than one blood meal, while about one-third of engorged females in Puerto Rico contained multiple blood meals [41]. Although the results of the present study cannot be explicitly compared with these earlier studies [41,47] because constraints of methodology applied; nevertheless, nearly one-quarter (38/145) of *Ae. aegypti* blood meals were mixed and accurately identified by next-generation sequencing technology. Females of *Ae. aegypti* have been shown to take blood meals 2–3 times per gonotrophic cycle, seeking and rapidly taking another meal, two to five hours after taking the first one [41,48]. This probably explains the high proportion of mixed dog-human feeding observed in the present study.

A plausible explanation for the low anthropophily observed in *Ae. aegypti* could be the relatively low human density compared to dogs. Only three of the households surveyed for the animal census were without a pet dog, and that was nullified by the number of stray dogs roaming the streets of the Pedro J. Méndez neighbourhood. This explanation is corroborated by the observation that *Cx. quinquefasciatus*, a species ubiquitously classified as ornithophilic [8, 44], also fed proportionally more on dogs than avian hosts. 304 (78.3%) *Cx. quinquefasciatus* out of 388 individuals examined fed on dogs (254 individuals contained single feeding with dog blood and 50 individuals contained mixed feeding with dog blood; Figs 2 and 3). Indeed, the relative availability of vertebrate hosts can profoundly influence the feeding patterns of host seeking mosquitoes [49]. Although, the estimated abundance of humans in the study area (14.4%) was higher that observed for dogs (11.8%; Table 2), it is significantly lower when compared with the relative abundance of all non-human hosts estimated. Host selection pattern in *Cx. quinquefasciatus* has been suggested to follow the domestic environment [8], which is expected to be predominantly non-human in this study.

Aedes aegypti evolved as an effective vector as part of forest-dwelling mosquito group with resting preferences on tree tops and breeding preferences in accumulated water in tree holes. Domestication of these species via the replacement of natural larval habitat with artificial container habitats in the urban landscape resulted in the adaptation to the human environment, and selected for highly endophilic and endophagic behaviour [41,50]. However, *Ae. aegypti* has spread throughout much of the tropical and subtropical regions of the world, with very unique urban landscapes and very different degrees of humans as viable hosts. For example, housing quality and characteristics are known to influence the ability of mosquitoes to enter

homes [15]. Additionally, continuous application of insecticides to households using ultra-low volume adulticides, likely further limits the endophilic and endophagic behaviour in certain regions. Although not explicitly investigated in the current study, fewer mosquitoes were collected indoors using the CDC backpack aspirators, as a proxy for indoor collection, than any of the other traps placed outside (S2 Table). Elizondo-Quiroga *et al.*[8] also observed a higher density of mosquitoes when collected outdoors than indoor in Nuevo Leon. Likewise, a two-year surveillance study of *Ae. aegypti* in Texas showed that outdoor collections in low-income neighbourhoods were almost eight times that of indoor collections, with this ratio increasing to as high as 15 times in middle-income communities [15]. A small change in host defensiveness has been predicted to drive a larger change in host choice [49]. Mosquito blood meal sources often occurring in aggregation are sometimes density-dependent, and anthropophagous insects tend to select for less defensive hosts, a host choice selection mechanism reinforced through limited learning [41,49]. Selection for host preference after insecticide application has been reported in *Anopheles gambiae*, with accompanying behavioural change from indoor to outdoor resting, and feeding on cattle and other non-human hosts [47].

The neurological behaviour underlying mosquito feeding preferences is not well understood, and few efforts have been made to understand the genetic basis [41,47]. It is not audacious to speculate that host selection and preference could be regulated by as yet unidentified genes, which have undergone positive selection in response to conditions that threaten survival.

The low incidence of human feeding of Ae. aegypti observed in the present study is significant with regards to disease transmission dynamics and control. In 1970, Hess & Hayes [51] proposed that the concept of zooprophylaxis, wherein non-human mammals available as hosts can reduce bites to humans by dengue vectors, thus reducing the arbovirus transmission potential. This occurrence has been demonstrated in Kenya and Uganda where non-human host feeding by Ae. aegypti was considered responsible for this species being a poor vector of vellow fever virus [52]. The suggestion to genetically modify vectors in order to eliminate the chemical signals that promote human-feeding preference has not been seriously considered due to the uncertainty of potential inherent risk [41], leaving zooprophylaxis as a safe and viable option. The utilization of non-human hosts and exophilic feeding observed in Ae. aegypti populations in Reynosa and in the South Texas study [43] suggests that zooprophylaxis could be a potential control measure to consider. In places where dogs contribute substantially to Ae. *aegypti* feeding, treating dogs with anti-parasitic drugs would not only protect them from ectoand endo-parasites, but could also reduce vector populations and reduce the risk of human exposure to arboviruses. For example, ivermectin has been reported to be a safe, short-term insecticide useful for malaria control in areas where mosquitoes are exophagic or exophilic [53]. In some cases, mosquito populations were reduced up to 80% after feeding on ivermectin treated individuals [54]. We also support our argument that anti-parasitic drugs can have an effect on vector populations in the context of the same vector/virus system. For instance, Deus et al. [55] found that ivermectin induces Ae. aegypti adult mortality and decreases the hatch rate of their eggs.

Diversion of vector bites to non-human competent hosts reduces viral amplification and reduces potentially infectious bites to humans by dengue vectors, thus could be used as a control tool to limit arbovirus transmission. Although an earlier study in a similar Reynosa neighborhood isolated CHIKV from patients with dengue-like symptoms [56], and co-circulation of DENV1-3 and West Nile virus [57], all pools of host-seeking *Ae. aegypti* were negative for the presence of arbovirus RNA (DENV 1–4, ZIKV, and CHIKV), and all pools of *Cx. quinquefasciatus* were negative for WNV RNA. The explosive arrival of viruses transmitted by mosquitoes, such as CHIKV and ZIKV, into the western hemisphere has demonstrated their rapid

ability to become endemic, with significant public health consequences. These viruses are of public health significance in Mexico, and are zoonotic with transmission cycles involving wild or domestic animals. Enzootic transmission cycles with wild mammal reservoirs can sustain virus populations in nature and initiate new foci of urban transmission through spill-over events. As *Ae. aegypti* is a vector of DENV, CHIKV, and ZIKV, while *Cx. quinquefasciatus* is known vector of WNV, we can hypothesize that domestic dogs can serve as a sentinel species for predicting virus activity and human disease risk in Northern Mexico, given the observation of high host utilization of dogs. Canine sero-surveillance has been suggested as a useful tool for public health risk assessment [58]. A study in France and Africa demonstrated the usefulness of dogs as a sentinel for studying WNV epidemiology and circulation [11], and similar studies have been conducted to study Chagas diseases in Texas and Northern Mexico [58,59]. Several studies have suggested dogs can be a sentinel for WNV [11]. Thus, although dogs may be unable to achieve the infectious viremia of arboviruses, seroconversion and neutralizing antibody production increases the opportunity of detecting low level arbovirus transmission.

Here, VC of *Ae. aegypti* for DENV ranged from 112 to 172 depending on temperatures (between 22 and 30°C, respectively). However, if h were 100% (*i.e.* all *Ae. aegypti* biting humans; 97% of *Ae. aegypti* feeding on humans was documented in Thailand and Puerto Rico) [41], VC would increase 2.5 times (ranging from 287 to 437, respectively). Hence, if the proportion of *Ae. aegypti* bloodmeals on humans were even higher than that reported here, the transmission of dengue viruses in this area would likely be much higher. Pooled mosquitoes would not have contained blooded individuals to avoid false positive results when modeling the VC due to undigested blood meal. The vector competence is the product of several factors, and should be varying with time. For example, the factor IM(t)/N_M(t), which represents the prevalence of dengue virus infection should be varying with time. We assumed that it is a constant due to the absence of such data. Similarly, the ratio of the number of new cases to the number of humans is also assumed to be constant. Some parameters that are unavailable were derived from literature, such as human biting rate, a, and the ratio of the number of mosquitoes to the number of humans are also estimated assuming that there are four people in one household. The survival rate varying with time is also from literature. The estimations would be more accurate with more data available.

In conclusion, the population of *Ae. aegypti* in Reynosa, Mexico fed on humans only 39.3% of the time, probably due to the abundance of non-human hosts in the residential neighbourhood and the low human density. The high rate of non-human blood meals for *Ae. aegypti* and dog blood meals for *Cx. quinquefasciatus* suggests that zooprophylaxis could be a control option to consider for DENV, ZIKV, CHIKV, and WNV, respectively. Nonetheless, the high number of blood meals from dogs and cats is in of itself a public health concern, due to the concomitant increase in the likelihood of zoonotic pathogen transmission (such as canine heartworm) and the potential spill-over to human populations. This study identifies the need to re-evaluate our knowledge of host utilization by *Ae. aegypti* and *Cx. quinquefasciatus* by meta-barcoding-like approach with next-generation deep sequencing technology in other regions of Mexico and the tropics in general; Determining VC's components would be useful determinants to help tailor successful vector control programs.

Supporting information

S1 Table. Vertebrate-specific primers used for first round PCR. The sequences of the M13F and M13R tails are in bold, while the COI-specific sequences are in regular font. The M13 tails served as second round PCR primer binding sites, on which Ion Torrent sequencing adapters and UMIs (IonXpress 1–96) were fused. (DOCX)

S2 Table. Collection data of mosquitoes collected in Reynosa in May-June and September-November 2018 for this study.

(DOCX)

S3 Table. Number of read counts for each host identification in individual *Aedes aegypti* and *Culex quinquefasciatus* examined through next-generation deep sequencing. (DOCX)

S4 Table. Host population densities documented by survey in the study area of Pedro J. Mendez neighbourhood, Reynosa, Northern Mexico. (DOCX)

Acknowledgments

We are grateful to the residents of the neighbourhood Pedro J. Mendez in Reynosa that granted permission to trap mosquitoes on their properties. We thank Lihua Wei, Janet J. Flores-García, Ricardo Palacios-Santana, Irma G. Cobos-Sosa, and Cristian G. Delgado-Corona for assistance in the field work. We thank Wendy Tang and Estelle Martin for assistance with testing mosquito pools for viruses. We thank the World Reference Center for Emerging Viruses and Arboviruses (WRCEVA) at the University of Texas Medical Branch and the Centers for Disease Control and Prevention for providing the dengue virus positive controls used in this study.

Author Contributions

Conceptualization: Paul D. N. Hebert, Gabriel L. Hamer, Mario A. Rodríguez-Pérez.

Data curation: Nadia A. Fernández-Santos, Adeniran A. Adebiyi, Sean W. J. Prosser.

- Formal analysis: Adeniran A. Adebiyi, Sean W. J. Prosser, Anthony R. Fooks, Gabriel L. Hamer, Ling Xue.
- Funding acquisition: Paul D. N. Hebert, Gabriel L. Hamer, Mario A. Rodríguez-Pérez.
- Investigation: José Guillermo Estrada-Franco, Nadia A. Fernández-Santos, Adeniran A. Adebiyi, María de J. López-López, Jesús A. Aguilar-Durán, Luis M. Hernández-Triana, Sean W. J. Prosser, Mario A. Rodríguez-Pérez.
- Methodology: José Guillermo Estrada-Franco, Nadia A. Fernández-Santos, Adeniran A. Adebiyi, María de J. López-López, Jesús A. Aguilar-Durán, Luis M. Hernández-Triana, Sean W. J. Prosser, Gabriel L. Hamer.

Project administration: Nadia A. Fernández-Santos.

Resources: Anthony R. Fooks.

Supervision: Nadia A. Fernández-Santos.

Validation: Sean W. J. Prosser, Ling Xue.

Visualization: Sean W. J. Prosser.

Writing - original draft: Adeniran A. Adebiyi, Mario A. Rodríguez-Pérez.

Writing – review & editing: Adeniran A. Adebiyi, Sean W. J. Prosser, Paul D. N. Hebert, Anthony R. Fooks, Gabriel L. Hamer, Mario A. Rodríguez-Pérez.

References

- 1. Burkett-Cadena ND. Mosquitoes of the southeastern United States2013. 1 online resource (203 pages) p.
- Scott TW, Clark GG, Lorenz LH, Amerasinghe PH, Reiter P, Edman JD. Detection of multiple blood feeding in Aedes aegypti (Diptera: Culicidae) during a single gonotrophic cycle using a histologic technique. Journal of medical entomology. 1993; 30(1):94–9. https://doi.org/10.1093/jmedent/30.1.94 PMID: 8433350.
- Logue K, Keven JB, Cannon MV, Reimer L, Siba P, Walker ED, et al. Unbiased Characterization of Anopheles Mosquito Blood Meals by Targeted High-Throughput Sequencing. PLoS neglected tropical diseases. 2016; 10(3):e0004512. https://doi.org/10.1371/journal.pntd.0004512 PMID: 26963245; PubMed Central PMCID: PMC4786206.
- Mukabana WR, Takken W, Knols BG. Analysis of arthropod bloodmeals using molecular genetic markers. Trends in parasitology. 2002; 18(11):505–9. <u>https://doi.org/10.1016/s1471-4922(02)02364-4</u> PMID: 12473367.
- Mwangangi JM, Mbogo CM, Nzovu JG, Githure JI, Yan G, Beier JC. Blood-meal analysis for anopheline mosquitoes sampled along the Kenyan coast. Journal of the American Mosquito Control Association. 2003; 19(4):371–5. PMID: 14710739.
- Tempelis CH. Host-feeding patterns of mosquitoes, with a review of advances in analysis of blood meals by serology. Journal of medical entomology. 1975; 11(6):635–53. <u>https://doi.org/10.1093/jmedent/11.6.635</u> PMID: 235647.
- Beier JC, Perkins PV, Wirtz RA, Koros J, Diggs D, Gargan TP 2nd, et al. Bloodmeal identification by direct enzyme-linked immunosorbent assay (ELISA), tested on Anopheles (Diptera: Culicidae) in Kenya. Journal of medical entomology. 1988; 25(1):9–16. https://doi.org/10.1093/jmedent/25.1.9 PMID: 3357176.
- Elizondo-Quiroga A, Flores-Suarez A, Elizondo-Quiroga D, Ponce-Garcia G, Blitvich BJ, Contreras-Cordero JF, et al. Host-feeding preference of Culex quinquefasciatus in Monterrey, northeastern Mexico. Journal of the American Mosquito Control Association. 2006; 22(4):654–61. https://doi.org/10. 2987/8756-971X(2006)22[654:HPOCQI]2.0.CO;2 PMID: 17304933.
- Reeves LE, Gillett-Kaufman JL, Kawahara AY, Kaufman PE. Barcoding blood meals: New vertebratespecific primer sets for assigning taxonomic identities to host DNA from mosquito blood meals. PLoS neglected tropical diseases. 2018; 12(8):e0006767. https://doi.org/10.1371/journal.pntd.0006767 PMID: 30161128; PubMed Central PMCID: PMC6135518.
- Lardeux F, Loayza P, Bouchite B, Chavez T. Host choice and human blood index of Anopheles pseudopunctipennis in a village of the Andean valleys of Bolivia. Malaria journal. 2007; 6:8. https://doi.org/10. 1186/1475-2875-6-8 PMID: 17241459; PubMed Central PMCID: PMC1783659.
- Davoust B, Leparc-Goffart I, Demoncheaux JP, Tine R, Diarra M, Trombini G, et al. Serologic surveillance for West Nile virus in dogs, Africa. Emerging infectious diseases. 2014; 20(8):1415–7. https://doi.org/10.3201/eid2008.130691 PMID: 25061842; PubMed Central PMCID: PMC4111180.
- Hebert PD, Cywinska A, Ball SL, deWaard JR. Biological identifications through DNA barcodes. Proceedings Biological sciences. 2003; 270(1512):313–21. <u>https://doi.org/10.1098/rspb.2002.2218</u> PMID: 12614582; PubMed Central PMCID: PMC1691236.
- Moran AJ, Prosser SWJ, Moran JA. DNA metabarcoding allows non-invasive identification of arthropod prey provisioned to nestling Rufous hummingbirds (Selasphorus rufus). PeerJ. 2019; 7:e6596. https:// doi.org/10.7717/peerj.6596 PMID: 30863689; PubMed Central PMCID: PMC6407503.
- Pappa V, Reddy M, Overgaard HJ, Abaga S, Caccone A. Estimation of the Human Blood Index in malaria mosquito vectors in Equatorial Guinea after indoor antivector interventions. The American journal of tropical medicine and hygiene. 2011; 84(2):298–301. <u>https://doi.org/10.4269/ajtmh.2011.10-0463</u> PMID: 21292902; PubMed Central PMCID: PMC3029185.
- Martin E, Medeiros MCI, Carbajal E, Valdez E, Juarez JG, Garcia-Luna S, et al. Surveillance of Aedes aegypti indoors and outdoors using Autocidal Gravid Ovitraps in South Texas during local transmission of Zika virus, 2016 to 2018. Acta tropica. 2019; 192:129–37. https://doi.org/10.1016/j.actatropica.2019. 02.006 PMID: 30763563.
- Darsie RF, Ward RA. Identification and geographical distribution of the mosquitoes of North America, north of Mexico. 2nd ed. Gainsville: University Press of Florida; 2005. xiv, 383 p. p. https://doi.org/10. 2987/8756-971X(2005)21[128:ROTPOC]2.0.CO;2 PMID: 16033112
- 17. Brugman V. Host selection and feeding preferences of farm-associated mosquitoes (Diptera: Culicidae) in the United Kingdom 2016.
- 18. Greenberg JA, Lujan DA, DiMenna MA, Wearing HJ, Hofkin BV. Identification of blood meal sources in Aedes vexans and Culex quinquefasciatus in Bernalillo County, New Mexico. Journal of insect science.

2013; 13:75. https://doi.org/10.1673/031.013.7501 PMID: 24224615; PubMed Central PMCID: PMC3835048.

- Verdonschot PFM, Besse-Lototskaya AA. Flight distance of mosquitoes (Culicidae): A metadata analysis to support the management of barrier zones around rewetted and newly constructed wetlands. Limnologica. 2014; 45:69–79. https://doi.org/10.1016/j.limno.2013.11.002.
- Ivanova NV, Dewaard JR, Hebert PDN. An inexpensive, automation-friendly protocol for recovering high-quality DNA. Molecular Ecology Notes. 2006; 6(4):998–1002. https://doi.org/10.1111/j.1471-8286. 2006.01428.x
- Martin M. Cutadapt removes adapter sequences from high-throughput sequencing reads. 2011. 2011; 17(1):3. Epub 2011-08-02. https://doi.org/10.14806/ej.17.1.200
- Edgar RC. UPARSE: highly accurate OTU sequences from microbial amplicon reads. Nature methods. 2013; 10(10):996–8. https://doi.org/10.1038/nmeth.2604 PMID: 23955772.
- 23. Alm E, Lesko B, Lindegren G, Ahlm C, Soderholm S, Falk KI, et al. Universal single-probe RT-PCR assay for diagnosis of dengue virus infections. PLoS neglected tropical diseases. 2014; 8(12):e3416. https://doi.org/10.1371/journal.pntd.0003416 PMID: 25522325; PubMed Central PMCID: PMC4270494.
- Lanciotti RS, Kosoy OL, Laven JJ, Panella AJ, Velez JO, Lambert AJ, et al. Chikungunya virus in US travelers returning from India, 2006. Emerging infectious diseases. 2007; 13(5):764–7. <u>https://doi.org/</u> 10.3201/eid1305.070015 PMID: 17553261; PubMed Central PMCID: PMC2738459.
- Lanciotti RS, Kosoy OL, Laven JJ, Velez JO, Lambert AJ, Johnson AJ, et al. Genetic and serologic properties of Zika virus associated with an epidemic, Yap State, Micronesia, 2007. Emerging infectious diseases. 2008; 14(8):1232–9. <u>https://doi.org/10.3201/eid1408.080287</u> PMID: <u>18680646</u>; PubMed Central PMCID: PMC2600394.
- Hamer GL, Walker ED, Brawn JD, Loss SR, Ruiz MO, Goldberg TL, et al. Rapid amplification of West Nile virus: the role of hatch-year birds. Vector borne and zoonotic diseases. 2008; 8(1):57–67. <u>https:// doi.org/10.1089/vbz.2007.0123 PMID: 18237262</u>.
- 27. Hess AD, Hayes RO, Tempelis CH. The use of the forage ratio technique in mosquito host preference studies. Mosquito News. 1968;28(3).
- Boreham PF, Garrett-Jones C. Prevalence of mixed blood meals and double feeding in a malaria vector (Anopheles sacharovi Favre). Bulletin of the World Health Organization. 1973; 48(5):605–14. PMID: 4544148; PubMed Central PMCID: PMC2482934.
- Garrett-Jones C. The Human Blood Index of Malaria Vectors in Relation to Epidemiological Assessment. Bulletin of the World Health Organization. 1964; 30:241–61. PMID: 14153413; PubMed Central PMCID: PMC2554803.
- Rodriguez-Perez MA, Fernandez-Santos NA, Orozco-Algarra ME, Rodriguez-Atanacio JA, Dominguez-Vazquez A, Rodriguez-Morales KB, et al. Elimination of Onchocerciasis from Mexico. PLoS neglected tropical diseases. 2015; 9(7):e0003922. https://doi.org/10.1371/journal.pntd.0003922 PMID: 26161558; PubMed Central PMCID: PMC4498594.
- Onyango MG, Bialosuknia SM, Payne AF, Mathias N, Kuo L, Vigneron A, et al. Increased temperatures reduce the vectorial capacity of Aedes mosquitoes for Zika virus. Emerging microbes & infections. 2020; 9(1):67–77. https://doi.org/10.1080/22221751.2019.1707125 PMID: 31894724.
- 32. Chan M, Johansson MA. The incubation periods of Dengue viruses. PloS one. 2012; 7(11):e50972. https://doi.org/10.1371/journal.pone.0050972 PMID: 23226436; PubMed Central PMCID: PMC3511440.
- Sallum MAM, Conn JE, Bergo ES, Laporta GZ, Chaves LSM, Bickersmith SA, et al. Vector competence, vectorial capacity of Nyssorhynchus darlingi and the basic reproduction number of Plasmodium vivax in agricultural settlements in the Amazonian Region of Brazil. Malaria journal. 2019; 18(1):117. https://doi.org/10.1186/s12936-019-2753-7 PMID: 30947726; PubMed Central PMCID: PMC6449965.
- Liu-Helmersson J, Stenlund H, Wilder-Smith A, Rocklov J. Vectorial capacity of Aedes aegypti: effects of temperature and implications for global dengue epidemic potential. PloS one. 2014; 9(3):e89783. https:// doi.org/10.1371/journal.pone.0089783 PMID: 24603439; PubMed Central PMCID: PMC3946027.
- Yang HM, Macoris ML, Galvani KC, Andrighetti MT, Wanderley DM. Assessing the effects of temperature on the population of Aedes aegypti, the vector of dengue. Epidemiology and infection. 2009; 137 (8):1188–202. https://doi.org/10.1017/S0950268809002040 PMID: 19192322.
- Rodriguez-Perez MA, Russell TL, Olguin-Rodriguez O, Laredo-Tiscareño SV, Garza-Hernández JA et al. Dengue serotypes circulating in Aedes aegypti and humans in a poor or peripheral neighborhood at Reynosa, Mexico. Southwestern Entomologist. 2020; 45 (4): 1025–1038.
- 37. Garza-Hernandez JA, Rodriguez-Perez MA, Salazar MI, Russell TL, Adeleke MA, de Luna-Santillana Ede J, et al. Vectorial capacity of Aedes aegypti for dengue virus type 2 is reduced with co-infection of Metarhizium anisopliae. PLoS neglected tropical diseases. 2013; 7(3):e2013. https://doi.org/10.1371/ journal.pntd.0002013 PMID: 23505581; PubMed Central PMCID: PMC3591344.

- Yasuno M, Tonn RJ. A study of biting habits of Aedes aegypti in Bangkok, Thailand. Bulletin of the World Health Organization. 1970; 43(2):319–25. PMID: <u>5312528</u>; PubMed Central PMCID: PMC2427649.
- Baak-Baak C. Human blood as the only source of Aedes aegypti in churches from Merida, Yucatan, Mexico. 2018.
- 40. Scott TW, Morrison AC, Lorenz LH, Clark GG, Strickman D, Kittayapong P, et al. Longitudinal studies of Aedes aegypti (Diptera: Culicidae) in Thailand and Puerto Rico: population dynamics. Journal of medical entomology. 2000; 37(1):77–88. https://doi.org/10.1603/0022-2585-37.1.77 PMID: 15218910.
- **41.** Edman JD. Fitness advantages in multiple blood-feeding: the Aedes aegypti example. Ecological aspects for application of genetically modified mosquitoes. 2003:63–74.
- Scott TW, Chow E, Strickman D, Kittayapong P, Wirtz RA, Lorenz LH, et al. Blood-feeding patterns of Aedes aegypti (Diptera: Culicidae) collected in a rural Thai village. Journal of medical entomology. 1993; 30(5):922–7. https://doi.org/10.1093/jmedent/30.5.922 PMID: 8254642.
- Olson MF, Ndeffo-Mbah ML, Juarez JG, Garcia-Luna S, Martin E, Borucki MK, et al. High Rate of Non-Human Feeding by Aedes aegypti Reduces Zika Virus Transmission in South Texas. Viruses. 2020; 12 (4). https://doi.org/10.3390/v12040453 PMID: 32316394; PubMed Central PMCID: PMC7232486.
- Garcia-Rejon JE, Blitvich BJ, Farfan-Ale JA, Lorono-Pino MA, Chi Chim WA, Flores-Flores LF, et al. Host-feeding preference of the mosquito, Culex quinquefasciatus, in Yucatan State, Mexico. Journal of insect science. 2010; 10:32. <u>https://doi.org/10.1673/031.010.3201</u> PMID: <u>20578953</u>; PubMed Central PMCID: PMC3014751.
- 45. Kent RJ, Crabtree MB, Miller BR. Transmission of West Nile virus by Culex quinquefasciatus say infected with Culex Flavivirus Izabal. PLoS neglected tropical diseases. 2010; 4(5):e671. https://doi.org/ 10.1371/journal.pntd.0000671 PMID: 20454569; PubMed Central PMCID: PMC2864301.
- 46. Gomes LA, Duarte R, Lima DC, Diniz BS, Serrao ML, Labarthe N. Comparison between precipitin and ELISA tests in the bloodmeal detection of Aedes aegypti (Linnaeus) and Aedes fluviatilis (Lutz) mosquitoes experimentally fed on feline, canine and human hosts. Memorias do Instituto Oswaldo Cruz. 2001; 96(5):693–5. https://doi.org/10.1590/s0074-02762001000500020 PMID: 11500773.
- Gillies MT. Selection for Host Preference in Anopheles Gambiae. Nature. 1964; 203:852–4. https://doi. org/10.1038/203852a0 PMID: 14204067.
- Jones JC, Pilitt DR. Blood-feeding behavior of adult Aedes aegypti mosquitoes. The Biological bulletin. 1973; 145(1):127–39. https://doi.org/10.2307/1540353 PMID: 4732534.
- 49. Kelly DW. Why are some people bitten more than others? Trends in parasitology. 2001; 17(12):578–81. https://doi.org/10.1016/s1471-4922(01)02116-x PMID: 11756041.
- Rodríguez-Pérez M, Howard A, Reyes-Villanueva F. Biological Control of Dengue Vectors. Integrated Pest Management and Pest Control—Current and Future Tactics 2012. p. 241–70.
- Hess AD, Hayes RO. Relative potentials of domestic animals for zooprophylaxis against mosquito vectors of encephalitis. The American journal of tropical medicine and hygiene. 1970; 19(2):327–34. https://doi.org/10.4269/ajtmh.1970.19.327 PMID: 5443080.
- McClelland GA, Weitz B. Serological Identification of the Natural Hosts of Aedes Aegypti (L.) and Some Other Mosquitoes (Diptera, Culicidae) Caught Resting in Vegetation in Kenya and Uganda. Annals of tropical medicine and parasitology. 1963; 57:214–24. https://doi.org/10.1080/00034983.1963. 11686176 PMID: 14042652.
- Chaccour C, Lines J, Whitty CJ. Effect of ivermectin on Anopheles gambiae mosquitoes fed on humans: the potential of oral insecticides in malaria control. The Journal of infectious diseases. 2010; 202 (1):113–6. https://doi.org/10.1086/653208 PMID: 20482251.
- Tanne JH. Antiparasite drug ivermectin cuts mosquito numbers by 80%. BMJ. 2011; 343:d4355. https://doi.org/10.1136/bmj.d4355
- 55. Deus KM, Saavedra-Rodriguez K, Butters MP, Black WCt, Foy BD. The effect of ivermectin in seven strains of Aedes aegypti (Diptera: Culicidae) including a genetically diverse laboratory strain and three permethrin resistant strains. Journal of medical entomology. 2012; 49(2):356–63. https://doi.org/10. 1603/me11164 PMID: 22493855; PubMed Central PMCID: PMC3942497.
- 56. Laredo-Tiscareno SV, Machain-Williams C, Rodriguez-Perez MA, Garza-Hernandez JA, Doria-Cobos GL, Cetina-Trejo RC, et al. Arbovirus Surveillance near the Mexico-U.S. Border: Isolation and Sequence Analysis of Chikungunya Virus from Patients with Dengue-like Symptoms in Reynosa, Tamaulipas. The American journal of tropical medicine and hygiene. 2018; 99(1):191–4. https://doi.org/ 10.4269/ajtmh.18-0117 PMID: 29761765; PubMed Central PMCID: PMC6085774.
- 57. Laredo-Tiscareno SV, Garza-Hernandez JA, Salazar MI, De Luna-Santillana EJ, Tangudu CS, Cetina-Trejo RC, et al. Surveillance for Flaviviruses Near the Mexico-U.S. Border: Co-circulation of Dengue Virus Serotypes 1, 2, and 3 and West Nile Virus in Tamaulipas, Northern Mexico, 2014–2016. The

American journal of tropical medicine and hygiene. 2018; 99(5):1308–17. https://doi.org/10.4269/ajtmh. 18-0426 PMID: 30226141; PubMed Central PMCID: PMC6221250.

- Tenney TD, Curtis-Robles R, Snowden KF, Hamer SA. Shelter dogs as sentinels for Trypanosoma cruzi transmission across Texas. Emerging infectious diseases. 2014; 20(8):1323–6. https://doi.org/10. 3201/eid2008.131843 PMID: 25062281; PubMed Central PMCID: PMC4111162.
- Arce-Fonseca M, Carrillo-Sanchez SC, Molina-Barrios RM, Martinez-Cruz M, Cedillo-Cobian JR, Henao-Diaz YA, et al. Seropositivity for Trypanosoma cruzi in domestic dogs from Sonora, Mexico. Infectious diseases of poverty. 2017; 6(1):120. https://doi.org/10.1186/s40249-017-0333-z PMID: 28870247; PubMed Central PMCID: PMC5584529.